MEG Management Committee meeting

25.03.2009, 14:00hrs, West Wing Seminar Room, CBU

Minutes
· Present:  Friedemann Pulvermuller, Yury Shtyrov, Rik Henson, Gary Chandler, Danny Mitchell, Oleg Korzyukov, Olaf Hauk, Rhodri Cusack, Maarten Van Casteren, Marie Smith, Rob Anthony, Georgie Morrill, Clare Dine.
1. Apologies for absences
None

2. Matters arising 
None
3. Minutes of the last meeting
Approved.
4. Computing








RA reported that the HP workstation has been moved into the user room. A normal desktop computer has been fitted in the lab in its place with access to the internet etc and RA will be setting up a shared MEG admin space so that things like the new excel log sheet can be updated in the lab and GM can then have access when needed for doing the monthly figures.
RH asked why the machines in the user room are slow. RA informed the committee that it’s due to the memory
 and computing will be purchasing additional memory for them which should increase the speed again.

Action:  RA to sort memory/logins on HP workstations in the user room.
5. Lab business and progress
Recurring HPI problems – use a fifth coil?
MvC has been using a new set of coils which seem to be working fine with Maxfilter.  The committee decided that when using the 5th coil with EEG it is best to fix this to the crown of the head and if this fails to work, then the top-centre of the forehead.
Approval of operator status for Lisa Brindley
The committee decided that it would be best to approve operator status at the MEG Management Committees so that the decision doesn’t fall upon one person.  It was also decided that it would be good to make sure the approved person operates at least once every 2-3 months to keep up to date and familiar with the set-up.
Phantom recordings data
OH will have the phantom recording data ready for the next meeting.  OH has emailed the issue regarding the 26Hz noise from the empty room recordings to Helsinki to ask for advice.
Action: OH to give progress update at next meeting.
Digitising EEG channels
RH discussed that it is a bit confusing at present with the EOG electrodes are being recorded as 61-64.  OK showed a diagram and explained to the committee how he has managed to get round this, but the electrodes will need re-labelling. ????
Other issues
OH wondered if it was worth emailing Neuromag/Helsinki about the issue with the Trigger values being at 40, asking them whether they think this is normal. ????
6. MEG Lab stats for April
GM showed the monthly lab figures to the committee.  The usage is quite low at 62.8% but GM informed the committee that it should increase dramatically in May as looking at the scheduler shows it is very well booked.  FP also informed the committee that we have been asked by the IMC to produce lab stats in the same format as MRI.  GM is in the process of discussing this with Marion and will produce a draft of the April stats in the MRI format.
Action: GM to liaise with Marion about current MRI stats spreadsheet.
7. Any other business
FP reported that the QQR outcome was in positive for MEG except the 1 band 4 MEG methods researcher post was denied. A new operator post was approved and can realistically be advertised later in 2009.
8. Date of next meeting
Mon 29th May in the WWSR.
�unclear





